\documentclass[12pt]{article}

\pagestyle{empty}
\setlength{\parindent}{0pt}
\newcommand{\answerspace}{\vfill}
\newcommand{\problemspace}{\bigskip}

\begin{document}

{\bf Math 165: Honors Calculus I}
\hfill\makebox[2.5in][r]{Name:\hrulefill}

{\bf Quiz 5}\quad {\sl October 7, 1999}

\bigskip

\begin{enumerate}

\item
Find the area between the graphs of $f(x) = x^3 - 3x^2 + 3x$ and $g(x)
= x^2 - 2x + 2$ on the interval $[0,3]$.

\answerspace

\problemspace

\item
Calculate the average value of $f(x) = \sqrt{x}$ on the interval $[1,4]$.

\answerspace

\newpage

\item
If $\pi$ denotes the area of the circular disk of radius~1, prove that
the area of a circular disk of radius $r$ is $\pi r^2$.

\answerspace

\end{enumerate}

\end{document}
